Author's response to reviews

Title: Can self-reported disability assessment behaviour of insurance physicians be explained? Applying the ASE model

Authors:

Antonius JM Schellart (ton.schellart@vumc.nl)
Romy Steenbeek (romy.steenbeek@tno.nl)
Henny PG Mulders (henny.mulders@uwv.nl)
Johannes R Anema (h.anema@vumc.nl)
Herman Kroneman (herman.kroneman@uwv.nl)
Jan JM Besseling (jan.besseling@tno.nl)

Version: 5 Date: 18 July 2011

Author's response to reviews: see over
Subject
Submission revised manuscript

Dear Sir/madam,

We hereby submit our final version of the paper “Can self-reported disability assessment behaviour of insurance physicians be explained? Applying the ASE model”.

Major revisions:
The abstract present in the manuscript matches the abstract in the submission system.
Introduction was changed into Background.
All references are cited in numerical order.
A section titled ‘Additional files’ was added with a title and description.
We refer to the additional files in the manuscript.

Minor revisions:
Highlighting/tracking and post-it style comments were removed.
The figures were cropped as closely as possible, according to the online figure guide.

We hope that the manuscript is now according to the journal style.

Thank you for accepting our paper.

On behalf of all authors,

Yours faithfully,

R. Steenbeek